Author's response to reviews

Title: Five recurrent BRCA1/2 mutations are responsible for cancer predisposition in the majority of Slovenian breast cancer families

Authors:

Mateja Krajc (mkrajc@onko-i.si)
Erik Teugels (eteugels@uzbrussel.be)
Janez Zgajnar (jzgajnar@onko-i.si)
Guido Goelen (guido.goelen@uzbrussel.be)
Nikola Besic (nbesic@onko-i.si)
Srdjan Novakovic (snovakovic@onko-i.si)
Marko Hocevar (mhocevar@onko-i.si)
Jacques De Grève (jacques.degrev@uzbrussel.be)

Version: 3 Date: 28 August 2008

Author's response to reviews:

Mateja Krajc, MD, MSc
Dept of Epidemiology and Cancer Registers
Familial Cancer Counseling and Testing
Institute of Oncology Ljubljana
Zaloška 2, 1000 Ljubljana, SLOVENIA
Tel: 00 386 1 5879 573, Fax: 00 386 1 5879 406
e-mail: mkrajc@onko-i.si

Editorial Office
BMC Medical Genetics
Joseph Dunckley
Assistant Editor
BMC-series journals
Ljubljana, 28.08.2008

Dear Mr. Joseph Dunckley,

We should kindly ask you to consider the enclosed revised manuscript “Five recurrent BRCA1/2 mutations are responsible for breast cancer predisposition in the majority of Slovenian families” for the publication in the BMC Medical Genetics.

We improved the manuscript as addressed in the comments made by the
reviewers. All the points were addresses in the manuscript as suggested.

We also went through the manuscript formatting checklist one more time, so the revised manuscript conforms to all of the points.

We look forward to your favorable reply,
On behalf of the authors,

Sincerely yours,

Mateja Krajc